Author's response to reviews

Title: Acute hepatic failure and multi-system organ failure secondary to replacement of the liver with metastatic melanoma.

Authors:

Gilaad G Kaplan (ggkaplan@ucalgary.ca)
Shaun AC Medlicott (shaun.medlicott@cls.ab.ca)
Bruce Culleton (bruce.culleton@calgaryhealthregion.ca)
Kevin B Laupland (kevin.laupland@calgaryhealthregion.ca)

Version: 5 Date: 13 June 2005

Author's response to reviews: see over
Faculty of Medicine  
Departments of Medicine, Critical Care Medicine, Community Health Sciences, and Pathology and Laboratory Medicine

June 13, 2005

Iratxe Puebla, Senior Editor  
BMC-series journals  
Middlesex House  
34-42 Cleveland Street  
London W1T 4LB  
Tel: +44 (0)20 7079 4808  
Fax: +44 (0)20 7580 1938 or +44 (0)20 7636 4296  
email: editorial@biomedcentral.com  
Web: www.biomedcentral.com

Re: Acute hepatic failure and multi-system organ failure secondary to replacement of the liver with metastatic melanoma.

Dear Iratxe:

We were please to find your letter informing us that the above named report will be accepted in BMC Cancer.

I have spoken again with the patient’s next of kin and she will provide written informed consent for this article to be published at your request. I will fax this to you shortly. We have made the formatting changes along the lines requested. However, we did have difficulty cropping the figures. Please let me know if the revised figures are acceptable.

Sincerely Yours,

Kevin B. Laupland MD MSc FRCPC  
Assistant Professor, Faculty of Medicine  
University of Calgary